Author's response to reviews

Title: Prevalence of female genital mutilation and its effect on women's health in Bale zone, Ethiopia: a cross-sectional study

Authors:

Daniel Bogale Mr (bogaleodo@gmail.com)
Desalegn Markos Mr (desalegnmarkos@gmail.com)
Muhammedawel Kaso Mr (muhammedawel@gmail.com)

Version: 2 Date: 14 August 2014

Author's response to reviews:

On the abstract part, introduction was changed to background, key words were omitted and the abstract was double spaced
The whole document was double spaced, page number and line number was inserted to the document
competing interest was edited, tables and figure legends were added as a heading